Author's response to reviews

**Title:** A Phase II Clinical Trial to assess the safety of Clonidine in Acute Organophosphate Poisoning

**Authors:**

Polwattage M.S Perera ([mark@sactrc.org](mailto:mark@sactrc.org))
Shaluka F Jayammani ([shalukajaya@yahoo.com](mailto:shalukajaya@yahoo.com))
Raja Hettiarachchi ([raga@sactrc.org](mailto:raga@sactrc.org))
Chandana Abeysinghe ([chandana@sactrc.org](mailto:chandana@sactrc.org))
Harendra Karunathilaka ([harendra@sactrc.org](mailto:harendra@sactrc.org))
Andrew H. Dawson ([adawson@sactrc.org](mailto:adawson@sactrc.org))
Nick Buckley ([n.buckley@unsw.edu.au](mailto:n.buckley@unsw.edu.au))

**Version:** 6  **Date:** 12 August 2009

**Author's response to reviews:**

Requested revisions were entirely minor grammatical wording changes - these have all been made in the attached version.

There was a query about mean mean arterial pressure in a figure label - this is correct but admittedly odd wording. We have changed that to "mean MAP" - with MAP defined in the Figure heading.